
Observing Report VEGA 2016.12.  1  1  
Fred, Orlagh (Calern) + Chris (Mountain)

Config for the December Run             B1           B2             B3  
------------------------------

 E1(P1)     E2(P2)     W1(P2)
S2(P5)     S1(P4)                 
                W2(P5)                

SUMMARY

Summary of observations: 
V66: HD37202  1pt (not great)

Summary of problems:  
Closed most of night for humidity.  Opened for about 1 hour.
Red camera not responding for ~10-15mins, just as we were ready for obs!

Summary of other:  
E2 OK.  CC was still there when opened, HUT not responding. Chris sorted it out. So E2 OK.

01h30 High humidity so waiting, everything is AOK, and coffee is soooo good. E2 seems to be OK
02h00 Fred broke his chair – sorry correction, Fred's chair broke.
03h00 Humidity oscillating around 90
05h00 Humidity 90-100%
07h00 Humidity is going down ~70% but some clouds.  
07h45 Looks like clearing and humidity is down
08h00 Opening, clear. CC still in way of E2.  Hut server not responding. Chris sorted it out. 
08h15 Ready to observe, but red camera not responding.

---------------------------------------------------

1.     V66 - BE SURVEY – ANTHONY  
E2 – Pop2 – B2     E1 – Pop1 – B1 

HD  37202  :  
08h35 HD37202.E2E1.2016.12.11.08.26  
           Sp Cal. D_R2656.2016.12.11.08.56

We tried the 2017 program HD62623, but no luck, no other bright stars on this triplet or another 
doublet at this time.  However, it is too low for bad seeing.  We tried.  

09h30 Close because of humidity
12h00 Closed up

END


